Author's response to reviews

Title: Systematic analysis of the clinical and biochemical characteristics of maternally inherited hypertension in Chinese Han families associated with mitochondrial genome mutations

Authors:

yuqi liu (ametufo980869@163.com)
chao zhu (zhuchao@163.com)
jie yang (yangjie@163.com)
tong yin (yintong@163.com)
jinliao gao (gaojinl@163.com)
zongbin li (lizongb@163.com)
qinghua ma (magingh@163.com)
yang li (liyang@163.com)
minxin guan (guanmx@163.com)
yundai chen (ametufo4290@gmail.com)
xueping wang (wangxq@163.com)

Version: 5 Date: 26 November 2014

Author's response to reviews: see over
The article was revised according to your suggestion.

1) Copyediting
Further consideration of your manuscript is conditional on improvement of the English used - please bear in mind that as we are a free-access publisher, we cannot bear the costs of copyediting English ourselves. Please ensure particular attention is paid to the abstract. You should have a native English speaking colleague help you with this, if possible, or you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/info/authors/authorfaqs#12.

Revised by editors from Edanz.

2) Consent to publish.
Consent for publication of individual patient data
For all manuscripts that include details, images, or videos relating to individual participants, written informed consent for the publication of these must be obtained from the participants (or their parent or legal guardian in the case of children under 16) and a statement to this effect should appear in the manuscript. If the participant has died, then consent for publication must be sought from the next of kin of the participant. You can use our consent form to obtain consent for publication from the participant(s), or a consent form from your own institution or region if you prefer. This documentation must be made available to Editors on request, and will be treated confidentially. In cases where images are entirely unidentifiable and there are no details on individuals reported within the manuscript, consent for publication of images may not be required. The final decision on whether consent to publish is required lies with the Editor.

Informed consent was obtained from all participating members, under protocols approved by the ethics committee of the Chinese PLA General Hospital.

3) Please include details of the ethics protocols in your ethics statement.

Provided.

4) Please include a Conclusion heading in the manuscript (Revised)

5) Please provide email address details for all authors on the title page of the manuscript. (provided)

6) Please reformat the abstract to comply with our instructions and move the competing interests to the appropriate position in the manuscript.

Revised.
7) Please rename the 'Introduction' to 'Background'
   Revised

8) Please include detailed contributions for all authors in the 'Authors' contributions' section.
   http://www.biomedcentral.com/bmcgenomics/authors/instructions/researcharticle.
   Revised.